Dopamine Augmented Rehabilitation in Stroke (DARS): a multicentre double-blind, randomised controlled trial of co-careldopa compared with placebo, in addition to routine NHS occupational and physical therapy, delivered early after stroke on functional recovery

Gary A Ford, 1* Bipin B Bhakta, 2† Alastair Cozens, 3 Bonnie Cundill, 4 Suzanne Hartley, 4 Ivana Holloway, 4 David Meads, 5 John Pearn, 2 Sharon Ruddock, 4 Catherine M Sackley, 6 Eirini-Christina Saloniki, 5 Gillian Santorelli, 4 Marion F Walker 7 and Amanda J Farrin 4

¹Oxford University Hospitals NHS Foundation Trust, University of Oxford, Oxford, UK ²Academic Department of Rehabilitation Medicine, Leeds Institute of Rheumatic and Musculoskeletal Medicine (LIRMM), University of Leeds, Leeds, UK ³NHS Grampian, Aberdeen, UK

⁴Clinical Trials Research Unit, Leeds Institute of Clinical Trials Research, University of Leeds, Leeds, UK

⁵Academic Unit of Health Economics, Leeds Institute of Health Sciences, University of Leeds, Leeds, UK

⁶Faculty of Life Science and Medicine, King's College London, London, UK ⁷Rehabilitation and Ageing, Queens Medical Centre, University of Nottingham, Nottingham, UK

*Corresponding author gary.ford@ouh.nhs.uk †In memoriam

Declared competing interests of authors: Gary A Ford received personal fees from Lundbeck Ltd, Boehringer Ingelheim, Pfizer and AstraZeneca, and grants and personal fees from Athersys outside the submitted work. Alastair Cozens had equity in Skene Software Ltd and other financial activity outside the submitted work for SiLCK Clinical Solutions Ltd. David Meads is a member of the Health Technology Assessment (HTA) Elective and Emergency Specialist Care (EESC) Panel. Catherine M Sackley is a member of Health Services and Delivery Research researcher-led board. Amanda J Farrin is a member of the HTA Clinical Evaluation and Trials Board and the HTA Commissioning Strategy Group.

Published July 2019 DOI: 10.3310/eme06050

Plain English summary

Dopamine Augmented Rehabilitation in Stroke (DARS)

Efficacy and Mechanism Evaluation 2019; Vol. 6: No. 5

DOI: 10.3310/eme06050

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

S troke has a huge impact, and more than one-third of affected people will have moderate or severe disability that has an impact on quality of life and self-care. Rehabilitation, in the form of routine NHS physical or occupational therapy, promotes recovery of mobility through relearning to use the affected body parts and/or learning to compensate with the less affected side. However, many people are not able to walk despite physiotherapy. Small studies have found that certain drugs that affect the learning process may improve functional recovery and that this improvement occurs when the drugs are given at the same time as the therapy session. The Dopamine Augmented Rehabilitation in Stroke trial was designed to measure if combining these drugs with routine therapy leads to an improvement in the ability to walk independently 8 weeks after a stroke.

We allocated at random 593 patients with a stroke who were unable to independently walk \geq 10 metres indoors to receive either a drug called co-careldopa or a placebo for 6 weeks, in addition to their routine therapy. We measured function and ability to walk at 8 weeks to see if the drug had an impact on recovery. We found that adding co-careldopa to routine physical and occupational therapy was not effective in improving walking following stroke. Further studies might consider using daily drug administration and more-intensive therapy. Future studies are needed to identify measures that identify new treatments to improve stroke recovery.

Efficacy and Mechanism Evaluation

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

EME programme

The Efficacy and Mechanism Evaluation (EME) programme was set up in 2008 as part of the National Institute for Health Research (NIHR) and the Medical Research Council (MRC) coordinated strategy for clinical trials. The EME programme is broadly aimed at supporting 'science driven' studies with an expectation of substantial health gain and aims to support excellent clinical science with an ultimate view to improving health or patient care.

Its remit includes evaluations of new treatments, including therapeutics (small molecule and biologic), psychological interventions, public health, diagnostics and medical devices. Treatments or interventions intended to prevent disease are also included.

The EME programme supports laboratory based or similar studies that are embedded within the main study if relevant to the remit of the EME programme. Studies that use validated surrogate markers as indicators of health outcome are also considered.

For more information about the EME programme please visit the website: http://www.nets.nihr.ac.uk/programmes/eme

This report

The research reported in this issue of the journal was funded by the EME programme as project number 08/43/61. The contractual start date was in January 2010. The final report began editorial review in November 2015 and was accepted for publication in February 2017. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, NETSCC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the EME programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Ford et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Honorary Professor, University of Manchester, and Senior Clinical Researcher and Associate Professor, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk